\documentclass[12pt]{article}
%\usepackage{amsmath,amssymb}
\usepackage{amsmath,amsthm,amsfonts,amssymb,amscd,mathtools}
\textheight=22cm \textwidth=16cm \hoffset=-13mm
\renewcommand{\baselinestretch}{1.2}
\usepackage{graphicx}
%\usepackage{showkeys}
%\renewcommand{\baselinestretch}{1.15}
\voffset=-2.0cm
 \normalsize
\usepackage[cp1251]{inputenc}  %% 1
\usepackage[T2B]{fontenc}      %% 2
%\usepackage[russian]{babel}    %% 3
\usepackage{color}


\input mathsoc.sty
\def\submitted{30.05.2020}

\title{\bf Online-centered Gaussian processes\\ and applications}
\author{
A.I. Nazarov\footnote{ St.Petersburg Department of Steklov Mathematical Institute of Russian Academy of Science, Fontanka 27, St.Petersburg, 191023, Russia, and St.Petersburg State University, Universitetskaya emb. 7-9, St.Petersburg, 199034, Russia; E-mail: nazarov@pdmi.ras.ru}\setcounter{footnote}{6}
\ and
Ya.Yu. Nikitin\footnote{St.Petersburg State University, Universitetskaya emb. 7-9, St.Petersburg, 199034, Russia, and National Research University -- Higher School of Economics, Soyuza Pechatnikov 16, St.Petersburg, 190008, Russia; E-mail: y.nikitin@spbu.ru
}
}
\date{}
%\bigskip


\begin{document}
\begin{thebibliography}{99}

%\bibitem{Ai} Ai, X., Li, W.V. Karhunen--Lo\`eve expansions for the $m$-th order detrended Brownian motion, Sci. China Math., {\bf 57} (2014), 2043--2052.

%\bibitem{Bah}  Bahadur, R.R.  Stochastic comparison of tests, Ann. Math. Stat., {\bf 31} (1960), N2, 276--295.

%\bibitem{BNO} Beghin, L., Nikitin, Ya., Orsingher, E. Exact small ball constants for some Gaussian processes under $L^2$-norm, J.  Math. Sci., {\bf 128} (2005), N1, 2493--2502.

\bibitem{BS10} 
Birman, M.S., Solomyak, M.Z. {\it Spectral theory of self-adjoint operators in Hilbert space}, 2nd ed., revised and extended. Lan', St.Petersburg, 2010 (Russian); English transl. of the 1st ed.: Mathematics and Its Applications. Soviet Series, {\bf 5}, Kluwer, Dordrecht etc. 1987.


\bibitem{BHS} 
Brown, A., Halmos, P.R., Shields, A.L. Ces\`aro operators, Acta Sci. Math. (Szeged) {\bf 26} (1965), N1-2, 125--137.

%\bibitem{Cai} Cairoli, R., Walsh, J.B. Stochastic integrals in the plane. Acta Math. {\bf 134} (1975), 111--183.

\bibitem{ChiKl}
Chigansky, P., Kleptsyna, M. Exact asymptotics in eigenproblems for fractional Brownian covariance operators. Stoch. Proc. Appl., {\bf 128} (2018), N6, 2007--2059.

%\bibitem{DPY} Deheuvels, P., Peccati, G., Yor, M. On quadratic functionals of the Brownian sheet and related processes, Stoch. Proc. Appl., {\bf 116} (2006), N3, 493--538.

%\bibitem{Deh} Deheuvels, P.  Karhunen--Lo\`eve expansions of mean-centered Wiener processes. In: {\it High Dimensional Probability} (Institute of Mathematical Statistics, 2006), 62--76.

%\bibitem{Dona} Donati-Martin, C., Yor, M. Fubini's theorem for double Wiener integrals and the variance of the Brownian path, Annales de l'IHP. Probab. et Stat., {\bf 27} (1991), N2, 181--200.
    
%\bibitem{Doz} Dozzi, M. {\it Stochastic processes with multidimensional parameter}, Pitman Research Notes in Math. {\bf 194}, Longman, NY, 1989.

%\bibitem{Dur} Durbin, J. Asymptotic Distributions of Some Statistics Based on the Bivariate Sample Distribution Function. In: {\it Nonparametric Techniques in Statistical Inference} (Cambridge University Press, Cambridge, 1970), 435--449.

%\bibitem{HN1} Henze, N., Nikitin, Y.Y. A new approach to goodness-of-fit testing based on the integrated empirical process, J. Nonparam. Statist. {\bf 12} (2000), N3, 391--416.

%\bibitem{HN2} Henze, N., Nikitin, Ya.Yu. Watson-type goodness-of-fit tests based on the integrated empirical process. Mathematical Methods of Statistics. {\bf 11} (2002), N2,  183--202.

\bibitem{Ka} Kamke, E. {\it Differentialgleichungen Losungsmethoden und Losungen}, vol. II. Chelsea Publishing Company, New York, 1947.

\bibitem{KNN} Karol', A., Nazarov, A., Nikitin, Ya. Small ball probabilities for Gaussian random fields and tensor products of compact operators, Trans. AMS, {\bf360} (2008), N3, 1443--1474.

\bibitem{Khm} Khmaladze, E.V., Martingale approach in the theory of goodness-of-fit tests. Theory of Probability \& Its Applications. {\bf 26} (1982), N2, 240--257.

\bibitem{Khm2} Khmaladze, E. Unitary transformations, empirical processes and distribution free testing. Bernoulli, {\bf 22} (2016), N1, 563--588.

\bibitem{KLB} Kleptsyna, M.L., Le Breton, A. A Cameron--Martin type formula for general Gaussian processes -- a filtering approach, Stochastics, {\bf 72} (2002), N3, 229--250.

%\bibitem{Koro} Korolyuk, V.S., Borovskich, Y.V. {\it Theory of $U$-statistics},  Springer Science \& Business Media, {\bf 273}, 2013.

%\bibitem{Kri} Krivyakova, E.N., Martynov, G.V., Tyurin, Y.N. On the $\omega^2$-statistic distribution in the multidimensional case, Teoriya Veroyatnostei i ee Primeneniya, {\bf 22} (1977), N2, 415--420 (Russian); English transl.: Theory Probab. Appl., {\bf 22} (1978), N2, 406--410.


\bibitem{KMP}
Kruglyak, N., Maligranda, L., Persson, L.-E., Structure of the Hardy operator related to Laguerre polynomials and the Euler differential equation, Revista Matem\'atica Complutense, {\bf 19} (2006), N2, 467--476.



%\bibitem{Lach} Lachal, A.  Bridges of certain Wiener integrals. Prediction properties, relation with polynomial interpolation and differential equations. Application to goodness-of-fit testing. In: {\it Bolyai Math. Studies X. Limit Theorems}, Balatonlelle (Hungary), 1999, Budapest, 1--51.

%\bibitem{Lif} Lifshits, M. {\it Lectures on Gaussian processes}.  Springer, Berlin, Heidelberg, 2012.

%\bibitem{Mart} Martynov, G.V. {\it Omega-square criteria}. Moscow, Nauka, 1978 (Russian).

\bibitem{N09} Nazarov, A.I. Exact $L_2$-small ball asymptotics of Gaussian processes and the spectrum of boundary-value problems, J. Theor. Probab., {\bf22} (2009), N3, 640--665.

\bibitem{N19} Nazarov, A.I., Spectral asymptotics for a class of integro-differential equations arising in the theory of fractional Gaussian processes, Preprint available at https://arxiv.org/abs/ 1908.10299. 30p. To appear in Comm. Contemp. Math.

\bibitem{N20} Nazarov, A.I., On a family of ordinary differential equations integrable in elementary functions, 2020, to appear in Math. Notes.

\bibitem{NaNi04} Nazarov, A.I., Nikitin, Ya.Yu. Exact small ball behavior of integrated Gaussian processes under $L_2$-norm and spectral asymptotics of boundary value problems. Probab. Theory Rel. Fields, {\bf 129} (2004), N4, 469--494.


\bibitem{NaNi20} Nazarov, A.I., Nikitin, Ya.Yu., Spectral equivalence of Gaussian random functions: operator approach, Preprint available at https://arxiv.org/abs/2005.05771. 14p. 

\bibitem{NikS} Nikitin Y., Sporysheva P. On asymptotic efficiency of tests of fit based on the Deheuvels empirical process. J. Math. Sci.  {\bf 159} (2009), 317--323.


%\bibitem{PecY} Peccati, G., Yor, M. Identities in law between quadratic functionals of bivariate Gaussian processes, through Fubini theorems and symmetric projections, Banach Center Publications, {\bf 72} (2006), 235--250.

%\bibitem{Pe} Petrova, Yu.P. Spectral asymptotics for problems with integral constraints, Mathematical Notes, {\bf 102} (2017), N3, 369--377.

\bibitem{PZ} Polyanin, A.D., Zaitsev, V.F., {\it Handbook of Exact Solutions for Ordinary Differential Equations}, CRC Press, Boca Raton, 2003 

%\bibitem{Rev} Revuz, D., Yor, M. {\it Continuous Martingales and Brownian Motion}. 3rd edition, Springer-Verlag, 1998.

%\bibitem{Ro} Rosenblatt, M. Remarks on a multivariate transformation, Ann. Math. Stat., {\bf 23} (1952), N3, 470--472.

\bibitem{SW} Shorack, G.R., Wellner, J.A.  {\it Empirical Processes with Applications to Statistics},  SIAM, 2009.

\bibitem{W}  Watson, G.S. Goodness-of-fit tests on a circle. Biometrika, {\bf 48} (1961), N 1/2, 109--114.


\end{thebibliography}

\PutMathSocLogo


\maketitle

\begin{abstract}
We establish various properties of the online centering of Gaussian processes and discuss their application to goodness-of-fit testing.
\end{abstract}


{\bf Keywords:} Gaussian processes, online centering, spectral equivalence

\medskip


The operation of online centering was introduced in \cite{KLB}. 
For a Gaussian process $X$ on $[0,1]$, we consider the process
$$
\widehat X(x)=X(x)-\frac 1x\int\limits_0^x X(t)\,dt.
$$

{\bf Proposition} (\cite[Example 4]{KLB}, \cite[Proposition 6.3]{KNN}). {\it The online centered Brownian motion is spectrally equivalent to the usual centered Brownian motion:}
\begin{equation}\label{eq:W}
\widehat W(x)\sim %B(x)\sim 
\overline {W\vphantom{^1}}(x):=W(x)- \int\limits_0^1 W(t)\,dt.
\end{equation}
%(the last relation is known long ago \cite{Dona}, see also \cite{BNO}).
In \cite{KLB} this fact was proved by the Laplace transform while in \cite{KNN} a direct calculation of the spectrum of $\widehat W$ was given.

\medskip



In this paper we demonstrate the operator nature of the relation (\ref{eq:W}) and show that a similar identity holds for the entire class of online centered Gaussian processes. 
\medskip

Let $A$ and $B$ be compact operators in a Hilbert space $H$. We call $A$ and $B$ {\it spectrally equivalent} and write $A\sim B$ if their non-zero eigenvalues coincide (with the multiplicities). A typical example of such operators is given by operator products $AB\sim BA$, see, e.g., \cite[Section 3.10]{BS10}.

Also we call two Gaussian random functions $X$ and $Y$ {\it spectrally equivalent} and write $X \sim Y$ if their covariance operators $K_X$ and $K_Y$ are spectrally equivalent. The notion of spectral equivalence was introduced in the recent paper \cite{NaNi20} though examples of such functions, both in univariate and in multivariate case, were known much earlier, see the references in \cite{NaNi20}. 
\medskip

We define some operators in $L_2(0,1)$:  operators of integration from the left and from the right
$$
(Tu)(x)=\int\limits_0^{x}u(t)\,dt, \qquad (T^*u)(x)=\int\limits_{x}^1 u(t)\,dt,
$$
the orthogonal projector onto the subspace of constants, the multiplication operator
$$
(Pu)(x)=\int\limits_0^1 u(t)\,dt, \qquad (Su)(x)=xu(x),
$$
and the operator of online centering
$$
(\widehat{\mathbb T} u)(x)=u(x)-\frac 1x\int\limits_0^x u(t)\,dt.
$$


{\bf Proposition 1}. The following identities hold:
\begin{equation}\label{eq:1}
\widehat{\mathbb T}^*\widehat{\mathbb T}=I-P; \qquad \widehat{\mathbb T}\,\widehat{\mathbb T}^*=I.
\end{equation}


This statement shows that operators $\widehat{\mathbb T}$ and $\widehat{\mathbb T}^*$ form, respectively, the left and the right shifts in $L_2(0,1)$. This fact was proved in \cite{BHS}, see also \cite[Theorem 1.1]{KMP}. We give here an elementary proof for the reader's convenience. We have
$$
\widehat{\mathbb T}^*\widehat{\mathbb T}u= %&\, 
(I-T^*S^{-1})(I-S^{-1}T)u,
$$
i.e.
$$
(\widehat{\mathbb T}^*\widehat{\mathbb T}u)(x)=u(x)-
\frac 1x\int\limits_0^xu(t)\,dt-\int\limits_x^1\frac {u(t)}t\,dt+\int\limits_x^1 \frac 1{t^2}\int\limits_0^tu(s)\,dsdt.
$$
Since
$$
\int\limits_x^1 \Big(\frac {u(t)}t-\frac 1{t^2}\int\limits_0^tu(s)\,ds\Big)dt= \Big(\frac 1t\int\limits_0^tu(s)\,ds\Big)\Big|_x^1 = 
\int\limits_0^1u(t)\,dt-\frac 1x\int\limits_0^xu(t)\,dt,
$$
we arrive at
$$
(\widehat{\mathbb T}^*\widehat{\mathbb T}u)(x)=u(x)-\int\limits_0^1u(t)\,dt =
((I-P)u)(x).
$$
On the other hand,
$$
(\widehat{\mathbb T}\,\widehat{\mathbb T}^* u)(x)=u(x)-
\frac 1x\int\limits_0^xu(t)\,dt-\int\limits_x^1\frac {u(t)}t\,dt+\frac 1x\int\limits_0^x \int\limits_t^1\frac {u(s)}s\,dsdt.
$$
Integration by parts annihilates three last terms, and the statement follows.
\hfill$\square$
\medskip


Now we can formulate our first main result.
\medskip

{\bf Theorem 1}. {\it For any zero mean-value Gaussian process ${\cal X}(x)$ on $[0,1]$, the online centered process $\widehat{\cal X}(x)$ is spectrally equivalent to the usual centered process:}
\begin{equation}\label{eq:X}
\widehat {\cal X}(x)\sim \overline {{\cal X}\vphantom{^1}}(x):={\cal X}(x)-\int\limits_0^1 {\cal X}(t)\,dt.
\end{equation}


{\it Proof}. It is easy to see that the covariance operators of the online centered and the usual centered process admit the representation
$$
K_{\widehat {\cal X}}=\widehat{\mathbb T}\,K_{\cal X}\widehat{\mathbb T}^*; \qquad K_{\overline {{\cal X}\vphantom{^1}}}=(I-P)K_{\cal X}(I-P).
$$
Therefore, we can write down the following chain: 
$$
%\aligned
\widehat{\mathbb T}\cdot \big[K_{\cal X}\widehat{\mathbb T}^*\big]\sim  %&\,  
\big[K_{\cal X}\widehat{\mathbb T}^*\big]\cdot \widehat{\mathbb T}\stackrel{\bullet}{=} K_{\cal X} (I-P)= \big[K_{\cal X}(I-P)\big]\cdot(I-P)\sim (I-P)\cdot \big[K_{\cal X} (I-P)\big],
%\endaligned
$$
(the equality ($\bullet$) follows from Proposition 1), and the statement follows.\hfill$\square$
\medskip

{\bf Remark 1}. Notice that the online centered Brownian motion is a {\it Green Gaussian process}, i.e. its covariance function is the Green function of a boundary value problem for an ordinary differential operator, see \cite[Proposition 6.3]{KNN}. Since the spectral theory of ODOs is well developed, this fact always helps a lot in search of the spectrum, see, e.g., \cite{NaNi04}, \cite{N09}. In contrast, for many Green Gaussian processes, the corresponding online centered process is NOT a Green Gaussian process. For instance, this is the case for the Brownian bridge. However, the equivalence (\ref{eq:X}) associates the spectrum of $\widehat B$ with that of $\overline {B\vphantom{^1}}$ which is known long ago, see \cite{W}.
\medskip


Another interesting example is related to the fractional Brownian motion (FBM) $W^H$ that is a zero mean-value Gaussian process with covariance function
$$
G_{W^H}(x,y)=\frac 12 \big(x^{2H}+y^{2H}-|x-y|^{2H}\big),\qquad x,y\in[0,1],
$$
(here $H\in(0,1)$ is the so-called Hurst index, the case $H = \frac 12$ corresponds to the standard Brownian motion). 
Using Theorem 1 we obtain
$$
\widehat {W^H}(x)\sim \overline {W^H\vphantom{^1}}(x).
$$

Notice that all fractional processes are not Green Gaussian processes, and their spectrum is not known exactly. However, recently the sharp spectral asymptotics for $\overline {W^H\vphantom{^1}}$ were obtained in \cite{N19} using a breakthrough approach of \cite{ChiKl}.
\medskip

Remark 1 generates a natural question, for which Green Gaussian processes corresponding online centered process is again a Green Gaussian process. A partial answer is given by the following theorem.
\medskip

{\bf Theorem 2}. {\it Let $X(x)$ be a zero mean-value Green Gaussian process on $[0,1]$. 
Denote by ${\cal X}(x)$ the left-integrated process
$$
{\cal X}(x)=\int\limits_0^x X(t)\,dt, \qquad x\in[0,1].
$$ 
Then the online centered process $\widehat{\cal X}(x)$ is also a Green Gaussian process.
}
\medskip

Before giving the proof we recall that the assumption of theorem means that the covariance function $G_X$ satisfies
\begin{equation}\label{eq:Green}
LG_X(\cdot,y)=\delta(\cdot-y); \qquad G_X(\cdot,y)\in Dom(L),
\end{equation}
where $L$ is a self-adjoint ordinary differential operator of order $2\ell$, 
$$
L\equiv
(-1)^{\ell}D^{\ell}\left(p_{\ell}(x)D^{\ell}\,\right)+
D^{\ell-1}\left(p_{\ell-1}(x)D^{\ell-1}\,\right)+\dots+p_0(x)
$$
(here $D$ stands for the differentiation operator, and $p_{\ell}(x)>0$) and the domain $Dom(L)$ is defined by $2\ell$ boundary conditions. In operator terms, (\ref{eq:Green}) can be written as
$LK_X=I$.\medskip


{\it Proof}. It is easy to see that the covariance operators of ${\cal X}$ and $\widehat{\cal X}$ admit the representation
$$
K_{\cal X}=TK_XT^*; \qquad
K_{\widehat {\cal X}}=\widehat{\mathbb T}\,TK_XT^*\widehat{\mathbb T}^*. \
$$

We begin with the identity
\begin{equation}\label{eq:2}
\widehat{\mathbb T}\,T=S^{-1}TS,
\end{equation}
which follows from a simple integration by parts formula
$$
\int\limits_0^x tf(t)\,dt=x\int\limits_0^x f(t)\,dt-\int\limits_0^x \int\limits_0^t f(s)\,dsdt.
$$
So, we obtain
$$
K_{\widehat {\cal X}}=S^{-1}TSK_XST^*S^{-1}.
$$
We invert the operator factors consequently and arrive at $\widehat{\cal L}K_{\widehat {\cal X}}=I$, where $\widehat{\cal L}$ is an ODO of order $2\ell+2$ given by 
$$
\widehat{\cal L}\equiv xDx^{-1}Lx^{-1}Dx.
$$
Since for every $u\in L_2(0,1)$ we have 
$$
(ST^*S^{-1}u)(1)=0,\qquad K_XST^*S^{-1}u\in  Dom(L),\qquad (K_{\widehat {\cal X}}u)(0)=0,
$$
the domain $Dom(\widehat{\cal L})$ is defined by $2\ell+2$ boundary conditions
$$
u(0)=0; \qquad x^{-1}Dxu\in Dom(L);\qquad (Lx^{-1}Dxu)(1)=0.
$$
Thus, the covariance function $G_{\widehat {\cal X}}$ satisfies
\begin{equation*}
\widehat{\cal L}G_{\widehat {\cal X}}(\cdot,y)=\delta(\cdot-y); \qquad G_{\widehat {\cal X}}(\cdot,y)\in Dom(\widehat{\cal L}),
\end{equation*}
 and the statement follows.\hfill$\square$
\medskip


{\bf Remark 2}. In the case where $L$ is an operator with constant coefficients, the fundamental system of solutions to the equation $\widehat{\cal L}u=\mu u$ can be written in terms of elementary functions, see \cite{N20}. We stress that this family of explicitly solvable ODEs is not included into classical handbooks \cite{Ka, PZ}.
\medskip


Next, by virtue of the Karhunen--Lo\`eve expansion, spectrally equivalent Gaussian functions have equally distributed $L_2$-norms. By Theorem 1, the following identity in law holds for any zero mean-value Gaussian process ${\cal X}(x)$ on $[0,1]$:
$$
\|\widehat {\cal X}\|^2_{L_2(0,1)}\stackrel{d}{=}\|\overline {{\cal X}\vphantom{^1}}\|^2_{L_2(0,1)}.
$$
However, we provide a much stronger statement.
\medskip

{\bf Theorem 3}. {\it Let $X(x)$ be arbitrary random process on $[0,1]$, square integrable a.s. Then} 
$$
\|\widehat {\cal X}\|^2_{L_2(0,1)}=\|\overline {{\cal X}\vphantom{^1}}\|^2_{L_2(0,1)} \qquad \text{a.s.}
$$


{\it Proof}. Using Proposition 1 we derive
$$
\aligned
\int\limits_0^1 \widehat {\cal X}^2(t)\,dt= &\ \int\limits_0^1 (\widehat{\mathbb T}{\cal X})^2(t)\,dt=
\int\limits_0^1 (\widehat{\mathbb T}^*\widehat{\mathbb T}{\cal X})(t)\cdot{\cal X}(t)\,dt\\
= &\ \int\limits_0^1 ((I-P){\cal X})(t)\cdot{\cal X}(t)\,dt=\int\limits_0^1 ((I-P){\cal X})^2(t)\,dt=
\int\limits_0^1 \overline {{\cal X}\vphantom{^1}}^2(t)\,dt,
\endaligned
$$
 and the statement follows.\hfill$\square$
\medskip

The results obtained above may have an unexpected application to nonparametrric statistics, namely to goodness-of-fit testing. Consider the classical empirical process built on a uniform sample on $[0,1]$
$$
\xi_n(t) = \sqrt{n}( F_n(t) - t), \, 0 \le t \le 1,
$$
where $F_n(t)$ is the empirical distribution function. The functionals of the empirical process are the famous nonparametric statistics such as Kolmogorov, Cram\'er--von Mises, Watson, Anderson--Darling statistics, and many others.

All them are used for goodness-of-fit testing. It is well known, see, e.g., the classical monograph \cite{SW} that the empirical process converges weakly in the Skorokhod space $D[0,1]$ to the Brownian bridge, and therefore the limiting distributions of statistics listed above coincide with these of Brownian bridge, and therefore are well studied.

In nonparametric statistics, the researchers are very interested in transformations of the empirical process and in functionals from such processes in the hope of finding new, more powerful or efficient tests for fit. The examples of such transformations are the extracting of the martingale part and more general constructions due to Khmaladze \cite{Khm}, \cite{Khm2} or the so-called Deheuvels empirical process \cite{NikS}.

The online centered empirical process has never been considered in this context. It has the form
$$
\widehat \xi_n(t) = \xi_n(t) - \frac{1}{t} \int\limits_0^t \xi_n(s)ds, \qquad 0 \le t \le 1.
$$
This process converges in Skorokhod space  to the process $\widehat{B}$ which is spectrally equivalent to the usual centered Brownian bridge $\overline{B\vphantom{^1}}$ by Theorem 1.

The corresponding $\omega^2$-type statistic $\int\limits_0^1 \widehat \xi_n^2 (t) dt$ not only has the same limiting distribution as $\|\overline{B\vphantom{^1}}\|^2_{L_2(0,1)}$ but equals the Watson statistics $\int\limits_0^1 \overline\xi{}_n^2 (t) dt$ almost surely by Theorem 3. However, it would be interesting to calculate and compare the local Bahadur asymptotic efficiency of other statistics based on $\widehat \xi_n$ against standard alternatives. 



 %\section{Conclusion}


\subsection*{Acknowledgement}

This work was supported by the Russian Foundation of Basic Research Grant 20-51-12004.

%\section{Disclosure statement}

%No potential conflict of interests was reported by the authors.


\small




\end{document}



But the process $\overline{B\vphantom{^1}}$ first studied by Watson \cite{W} is well known and relatively simple. It was proved a long time ago, see \cite[\S 3.8]{SW}, that there exists the following equality in distribution
$$
\|\overline{B\vphantom{^1}}\|_{L_2(0,1)}\stackrel{d}{=}\pi ^{-1}\sup_{0\leq t\leq 1}|B(t)|.
$$
The random variable $\sup_{0\leq t\leq 1}|B(t)|$\ has the well-known
Kolmogorov distribution function, for which there exist explicit formulae
suitable for small and large values of the argument

Similar theory can be developed for the integrated empirical process, see \cite{HN1}, \cite{HN2}. Consider the process 
$$
\eta_n(t) = \int\limits_0^t \xi_n(s) ds,
$$
and, similarly to $\widehat \xi_n(t),$ we introduce the online centered integrated empirical process
$$
\widehat \eta_n(t) = \eta_n(t) - \frac{1}{t} \int\limits_0^t \eta_n(s) ds.
$$
Let $B^{[0]}$ denote the integrated Brownian bridge.  Then
we may expect that the limiting process is $ B^{[0]}(t) -\frac{1}{t}\int\limits_0^t B^{[0]}(s) ds$.
But according to Theorem 1
$$
B^{[0]}(t) -\frac{1}{t}\int\limits_0^t B^{[0]}(s) ds \sim  B^{[0]}(t) - \int\limits_0^1 B^{[0]}(s) ds. 
$$
The latter process was studied in \cite{HN2}. We know its covariance and its spectrum. %$\lambda_n = (\pi n)^{-4}.$
Therefore we also can numerically calculate the limiting distribution of $\int\limits_0^1 \widehat \eta_n^2(t)dt$ and the local approximate Bahadur efficiency of the corresponding goodness-of-fit test.
